Author's response to reviews

Title: The quality of clinical practice guidelines in traditional medicine in Korea: appraisal using the AGREE II instrument

Authors:

Tae-Young Choi (superoung@gmail.com)
Jiae Choi (choijiae0723@gmail.com)
Ju Ah Lee (motoong@kiom.re.kr)
Ji Hee Jun (zhixi04@kiom.re.kr)
Bongki Park (sanisapa@hanmail.net)
Myeong Soo Lee (drmslee@gmail.com)

Version: 6 Date: 22 June 2015

Author's response to reviews:

Dear Reviewers,

RE: The quality of clinical practice guidelines in traditional medicine in Korea: appraisal using the AGREE II instrument

On the behalf of my co-author, I would like to thank you for arranging peer-review of our manuscript and for your invitation to submit a revised version. We appreciate the effort of the reviewers, and believe that their constructive suggestions have resulted in a stronger manuscript for the Implementation Science’ readers.

Reviewer #1:

Comment 1)

There is one additional issue that needs to be corrected. In the flow chart, ?Database searching? should be ?Database Searches?.

Revised> We have revised it in the flow chart.

Thank you for your valuable comments.